Author’s response to reviews

Title: Clinical ethics dilemmas in a low-income setting - A national survey among physicians in Ethiopia

Authors:

Ingrid Miljeteig (Ingrid.Miljeteig@uib.no)
Frehiwot Defaye (frehiwot.berhane@gmail.com)
Dawit Desalegn (mdawitd@gmail.com)
Marion Danis (mdanis@cc.nih.gov)

Version: 2 Date: 24 Aug 2019

Author’s response to reviews:

Bergen 24.08.2019

Dear Editor of BMC Medical Ethics

Thank you so much for supporting the publication of our paper: “Clinical ethics dilemmas in a low-income setting - A national survey among physicians in Ethiopia”. We are very proud of how the paper looks like now, and are so grateful for the comments by the reviewers. We have accepted all the suggestions from reviewer 1 and have done the editorial changes you asked us to do. We have also uploaded the part of the questionnaire that was used for this part of our study. We hope all is ok now for publication!

Kind regards,

Ingrid Miljeteig, on behalf of the co-authors